Author's response to reviews

Title: Dietary factors associated with Metabolic Syndrome in Brazilian adults

Authors:

Erick P de Oliveira (erick_po@yahoo.com.br)
Kátia C Portero-McLellan (kmclellan@fmb.unesp.br)
Liciana V A Silveira (liciana@ibb.unesp.br)
Roberto C Burini (burini@fmb.unesp.br)

Version: 4 Date: 20 February 2012

Author's response to reviews: see over
Dear Sr.,

Please consider the enclosed manuscript entitled “Dietary factors associated with Metabolic Syndrome in Brazilian adults” with corrections as directed by the reviewer.

The corrections suggested by the reviewers are shaded in yellow in the body of the text:

- All the suggestions were changed according to suggestions of the reviewers.

- The categories of variables marital status and educational level were corrected in the methodology. There are two groups of comparison, so in the table 1 the percentage of one of one of these groups was described and the percentage of the other group is complementary.

Regards,

Erick Prado de Oliveira
Nutritional and Exercise Metabolism Center (CeMENutri). Department of Public Health, Botucatu Medical School, UNESP - Sao Paulo State of University, Brazil
Phone or Fax number: +55 (14) 3811-6128.
E-mail: erick_po@yahoo.com.br